Author’s response to reviews

Title: SKIP (Supporting Kids with diabetes In Physical activity): Feasibility of a randomised controlled trial of a digital intervention for 9-12 year olds with type 1 diabetes mellitus.

Authors:

Emily Knox (emily_knox2@hotmail.co.uk; emily.knox@nottingham.ac.uk)

Cris Glazebrook (cris.glazebrook@nottingham.ac.uk)

Tabitha Randell (tabitha.randell@nuh.nhs.uk)

Paul Leighton (paul.leighton@nottingham.ac.uk)

Boliang Guo (boliang.guo@nottingham.ac.uk)

James Greening (james.greening@uhl-tr.nhs.uk)

E. Bethan Davies (bethan.davies@nottingham.ac.uk)

Lori Amor (lori.amor@nottingham.ac.uk)

Holly Blake (Holly.Blake@nottingham.ac.uk)

Version: 2 Date: 08 Mar 2019

Author’s response to reviews:

Dear Editor,

Many thanks for accepting this article subject to some minor revisions. These have been completed and the final revised version attached. We have slightly revised the title to avoid confusion since the funded project is called SKIP, but the intervention itself is called STAK-D. The intervention involves website AND wearable device and therefore we have described this as a digital intervention to ensure both elements are included in the title which previously referred only to website.

We have also added an additional line to further clarify the clinical data outcome in the feasibility section and abstract.

kind regards,

Holly Blake (corresponding author)